Author's response to reviews

Title: Legionella spp. and legionellosis in Southeastern Italy: disease epidemiology and environmental surveillance in community and health care facilities

Authors:

Christian Napoli (c.napoli@igiene.uniba.it)
Fabrizio Fasano (c.napoli@igiene.uniba.it)
Roberta Iatta (c.napoli@igiene.uniba.it)
Giovanna Barbuti (c.napoli@igiene.uniba.it)
Teresa Cuna (c.napoli@igiene.uniba.it)
Maria Teresa Montagna (c.napoli@igiene.uniba.it)

Version: 8 Date: 27 October 2010

Author's response to reviews:

Dear Editorial Team,

thank you for your support, we made all changes required, included the submission of the figures without this protection and the division of "Health-care" and "Community" cells up into the correct number of rows, in tables 1-4.

I do apologize for this problem!

best regards

christian napoli